Author's response to reviews

Title: Determinants of unmet need for family planning among currently married women in Dangila town administration, Awi Zone, Amhara regional state; A cross sectional study.

Authors:

Ewunetu Genet (genetewnetu2012@gmail.com)
Gedefaw Abeje Fekadu (abejegedefaw@gmail.com)
Tadese Ejigu (tade_et@yahoo.com)

Version: 2 Date: 9 October 2014

Dear Editor,

We have made some revisions on the manuscript. The revision includes language and grammar correction. We also gave line numbers. Changes in the manuscript are highlighted yellow.

kind regards,